Author's response to reviews

Title: Synchronous well differentiated neuroendocrine tumour and gastrointestinal stromal tumour of the stomach: a case report.

Authors:

Vassilis D Samaras (vassilissamaras@yahoo.gr)
Periklis G Foukas (pfoukas@yahoo.com)
Konstantinos Triantafyllou (ktriant@med.uoa.gr)
Vassilia Leontara (vassilissamaras@gmail.com)
Dimitrios Tsapralis (tsapralis@yahoo.gr)
Eirini M Tsompanidi (tsompanidi@yahoo.gr)
Anastasios Machairas (anmach@med.uoa.gr)
Ioannis G Panayiotides (histodept@attikonhospital.gr)

Version: 4 Date: 16 March 2011

Author's response to reviews: see over
Athens, March 16, 2011

To: The Editor, BMC Gastroenterology

Re: Revised manuscript submission

Dear Dr. Shipley,

We have the pleasure to submit our revised manuscript entitled: ‘Synchronous well differentiated neuroendocrine tumour and gastrointestinal stromal tumour of the stomach: a case report.’ to your distinguished journal for publication.

In this third revised version of our manuscript we have performed all the necessary changes according to editorial requests.

Specifically:

**RESPONSE TO THE EDITORIAL REQUESTS**

1. **<Affiliations: Each affiliation should.....within an institution>>:** We have performed all the necessary changes according to your instructions [see: page 1 of “MANUSCRIPT”].

2. **<<Email address: please include.....the following format>>:** We have modified the format of email addresses based on your comments [see: page 1 of “MANUSCRIPT”].

We are looking forward for your response,

Yours sincerely,
Vassilis Samaras, MD, PhD

2nd Department of Pathology, National and Kapodistrian University of Athens, Medical School, “Attikon” University Hospital, Athens, Greece.

1 Rimini, 12462, Chaidari, Athens, Greece.

Tel: +30 (210) 5831045

Fax: +30 (210) 5831053

E-mail: vassilissamaras@yahoo.gr

vassilissamaras@gmail.com